Author's response to reviews

**Title:** Acute lethal intestinal gas gangrene of a 12 months old child after long-standing artificial nutrition as a cause of Pneumatosis coli: a case report

**Authors:**

Stefan S Kircher (stefan.kircher@gmx.de)
Rupert R Woessner (woessner_r@kinderklinik.uni-wuerzburg.de)
Hans-Konrad HK Mueller-Hermelink (path062@mail.uni-wuerzburg.de)
Hans-Ullrich HU Voelker (hans-ullrich.voelker@t-online.de)

**Version:** 2  **Date:** 8 November 2007

**Author's response to reviews:**

We changed the title in: Acute lethal intestinal gas gangrene of a 12 months old child after long-standing artificial nutrition as a cause of Pneumatosis coli: a case report.

The Abstract was included on page 2, composed of Introduction, Case presentation and Conclusion. The declaration of consent was separated from the acknowledgement.